Notes of the RICH1 Phone meeting 17/03/2006
People Present:
Bristol:  Richard
CERN: Carmelo, Christophe, Dave, Fabio, Nick, Olav, Werner 
IC:        Bill, Gabriel, Trevor
Oxford: Rohan(part)
RAL:     Peter, Derrick
Apologies for absence:
 Oxford: Neville, Tony    
 RAL: Mike    
Discussed Agenda Topics:
1. Gas Enclosure 
1.1 Transport to CERN:  Has arrived at CERN last Wednesday 8 March06. It was brought to building 156 and was mounted on its trolley. It is in a dirty state. Cleaning should be completed by Wednesday next week.  Then the GE will be put in the tent for leak testing.  After completing the leak tests it will be moved outside the tent for cleaning its inside.
1.2 Mounts: The mounts for the installation of the GE in the magnetic shield enclosure are being prepared at IC and should be ready by the end of next week. 

The parts to be clamped to the GE to prevent an accidental movement of the GE once installed in the pit will be manufactured only when the GE final position has been measured. These parts can be fitted later on.
1.3 Installation Hoist: All has been ordered. The delivery of the materials is expected for the end of next week.  Transport to CERN will be in April06.
1.4 Painting:   Some polyurethane paint samples have been ordered and should arrive next week. Akzo-Nobel in the UK have replied. They say that the epoxy paint that appeared to have ideal properties for us is no more in production and propose a   polyurethane paint instead. They will send the paint data sheet. 
The painting will be done after Easter.

1.5 PPSPS
The latest version is on EDMS. There are some errors in the calculations which will be corrected. Rohan will send the new version next week to Bill, which will then be available on EDMS.
1.6 Components
Aerogel mechanics are ready. Plan to do a test assembly in 156. 

All components/mechanics will be removed from the GE before its installation in the pit
2. VELO Seal
It is packed up in its crate.   Doing some modeling and preparing some drawings for its support/installation parts. The parts will be made and all will be sent to CERN in ~1month.
3. Exit Window
All components are ready. Have to get the test rig for the diaphragm up and running. The problem of filling the mould has been solved: bubble free mould. 

Exact figures for the opacity of the black silicone are not available. There is some light transmission when the sample is held in front of a light source. A sample will be sent to CERN for opacity testing. More trials to improve the opacity will be made. This is done by increasing the quantity black material in the silicone. Up to 50% black material can be added without compromising the silicone structural integrity. For the sample 20% black material has been added.
If the opacity tests fail a separate opaque material may need to be will be used.  
4. Beampipe Installation and Tooling
Two separate installation support mechanics will be used: one for the beampipe installation and another for the Exit Window. Trevor and Delio will meet to discuss (action Trevor).
5. Spherical Mirrors
CMA sent the monthly report. The CMA samples should be ready by the end of March06. The geometry is wrong. The mirrors are rotated by 90(.  CMA propose to use  materials for the demonstration mirror which are different form the final RICH1 mirrors, in contrast to what specified in the contract. CMA has been contacted immediately asking them to not start the production of the demonstration mirror.  There are several questions which need to be answered and there is a general concern in the moderate progress made by CMA. 
A phone meeting CMA will be organized next week.
Irradiation time for the CMA samples will be booked for the second half of April06 (action Fabio).
6. Flat mirrors
Milan should bring 20 flat mirrors to CERN next week. The mirrors will be measured, the reflective coating will be applied and then the mounts will be glued. Float glass will be used for the glue testing. 
Retainer Ring:  The polycarbonate discs using the material from IC have been machined. Polycarbonate material for the final retainer rings has been ordered.
Mechanics:  Have achieved a small improvement with the machined dowels. The limit of accuracy has been reached for the current design.  Maybe the design needs to be modified. The angle adjuster has not yet been tested because of the problem with the plate. The mirror box design from Richard still needs to be looked at (action Geoff). 
7. HPD Assembly
Meeting between people at IC and Oxford held at IC last week to discuss maintenance rails for HPD assembly. 
Cooling Plates: Sketch for the cooling plates system has been drawn. The materials for the columns will be ordered, cut to the needed dimensions, from the CERN stores and an Austrian company. The Genoa group said that the best time to make the cooling plates is in Aug-Sept06.   
8.  Aerogel
Aerogel mechanics (rails) are ready. Get drawings from Roberto to put in the model (action Trevor).
9.  Quartz Windows
The faulty (splinter) RICH2 quartz window can not be cut and used for RICH1.  

Once the quartz window is installed special care will have to be taken to avoid damaging it during the HPD installation. The risks for the beampipe during installation of the upper quartz window have been understood and minimized.
10.  LAMS

System already designed in 2004. The laser is at RAL. Paul Soler will order radiation hard fibers for both RICH1 and RICH2. A radiation tolerant CID camera has been ordered to test for RICH1. The camera control units for RICH1 will be placed on the balcony. Ken Wyllie has been informed of the cabling requirements.

11.  Magnetic Distortions

The Syracuse system which allows an absolute positioning measurement requires an optical fiber x-y transport stage in front of the HPDs. The allocated volume in the photon funnel for the Syracuse system is difficult to accommodate. Phone meeting planned for next Friday with Syracuse to discuss things.
12. Next meeting
Friday  31st of March 2006 at 10:00(GMT) 11:00(CET). 
